Author’s response to reviews

Title: Strategies for Recruitment and Retention of Underrepresented Populations with Chronic Obstructive Pulmonary Disease for a Clinical Trial

Authors:

David Thom (David.Thom@ucsf.edu)
Denise DeVore (Joanna.Devore@ucsf.edu)
Chris Chirinos (Chris.Chirinos@ucsf.edu)
Jessica Wolf (jnwolf1@gmail.com)
Devon Low (devonsrooster@gmail.com)
Rachel Willard-Grace (rachel.willard@ucsf.edu)
Stephanie Tsao (stephanie.tsao@sfdph.org)
Chris Garvey (chris.garvey@ucsf.edu)
Doranne Donesky (doranne.donesky@ucsf.edu)
George Su (george.su@ucsf.edu)
David Thom (dthom@stanford.eu)

Version: 2 Date: 31 Jan 2019

Author’s response to reviews:

We have made the requested changes to our manuscript with the exception of number 4; we are not able to identify a section “Economic, social and ethical factors affecting the implementation of research.”

Please let us know if there is anything else you need.
Regards,

David Thom, MD, PhD